%% @texfile{
%%     filename="TorusRHP.tex",
%%     version="1.1",
%%     date="July-2022",
%%     cdate="20201012",
%%     filetype="LaTeX2e",
%%     journal="Anal. Math. Phys. 12, 112 (2022)",
%%     copyright="Copyright (C) M. Piorkowski and G.Teschl",
%%     doi="10.1007/s13324-022-00715-4"
%%     }

\documentclass{amsart}
\usepackage{hyperref}
\usepackage{graphicx}
\usepackage{curves}
\usepackage{tikz}
\usetikzlibrary{arrows}
%\usepackage{showkeys}
%\usepackage{refcheck}
\usepackage{enumerate}
\usepackage{float}
\usepackage{comment}
\unitlength1cm

%%%%%%%%%
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{arXiv:#1}}
\newcommand*{\mailto}[1]{\href{mailto:#1}{\nolinkurl{#1}}}

%%%%%%%%%THEOREMS%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{hypothesis}[theorem]{Hypothesis}
\newtheorem{proposition}[theorem]{Proposition}

%%%%%%%%%%%%%%FONTS%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\R}{\mathbb{R}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\T}{\mathbb{T}}
%%%%%%%%%%%%%%%%%%ABBRS%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\nn}{\nonumber}
\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}
\newcommand{\bea}{\begin{eqnarray}}
\newcommand{\eea}{\end{eqnarray}}
\newcommand{\ul}{\underline}
\newcommand{\ol}{\overline}
\newcommand{\pa}{\partial}
\newcommand{\ti}{\tilde}
\newcommand{\wti}{\widetilde}
\newcommand{\wha}{\widehat}
\newcommand{\norm}[1]{\lVert#1 \rVert}
\newcommand{\abs}[1]{\lvert#1 \rvert}


\newcommand{\spr}[2]{\langle #1 , #2 \rangle}
\newcommand{\id}{\mathbb{I}}
\newcommand{\I}{\mathrm{i}}
\newcommand{\E}{\mathrm{e}}

\newcommand{\re}{\mathop{\mathrm{Re}}}
\newcommand{\im}{\mathop{\mathrm{Im}}}
\newcommand{\tr}{\mathop{\mathrm{tr}}}
\newcommand{\hatb}{{\hspace{-0.1cm}\bf{\Hat{\hspace{+0.1cm} b}}}}

\DeclareMathOperator{\res}{Res}
\DeclareMathOperator{\dn}{dn}
\DeclareMathOperator{\dom}{dom}
\newcommand{\db}{\mathfrak{D}}
\newcommand{\FOm}{\mathcal{F}(\ti{\Lambda})}
\DeclareMathOperator{\sgn}{sgn}
\DeclareMathOperator{\Cov}{Cov}

%%%%%%%%%%%%%%%GREEK%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\eps}{\varepsilon}
\newcommand{\vphi}{\varphi}
\newcommand{\si}{\sigma}
\newcommand{\la}{\lambda}
\newcommand{\ga}{\gamma}
\newcommand{\om}{\omega}

%%%%%%%%%%%%%%%%%%%%%%%%NUMBERING%%%%%%%%%%%%%%%%%%%%

\renewcommand{\labelenumi}{(\roman{enumi})}
\numberwithin{equation}{section}


%%%%%%
\newcommand{\sigI}{\begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}}
\newcommand{\rI}{\begin{pmatrix}  1 & 1 \end{pmatrix}}
\newcommand{\rN}{\begin{pmatrix}  0 & 0 \end{pmatrix}}

 \newcommand{\noprint}[1]{}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document}
\begin{thebibliography}{XXX}
\bibitem{BBEIM} E. Belokolos, A. Bobenko, V. Enol'skii, A. Its and V. Matveev, \textit{Algebro-Geometric Approach to Nonlinear Integrable Equations}, Springer Series in Nonlinear Dynamics, Berlin, 1994.
%%%%%
\bibitem{BKT16} M. Bertola, A. Katsevich and A. Tovbis, \textit{Singular Value Decomposition
of a Finite Hilbert Transform Defined on Several Intervals and the Interior Problem of Tomography:
The Riemann-Hilbert Problem Approach
}, Comm. Pure Appl. Math. {\bf 69}, 407--477 (2016).
%%%%%
\bibitem{Byrd} P. F. Byrd and M. D. Friedman, {\em Handbook of Elliptic Integrals for Engineers and Physicists}, Springer, Berlin, 1954.
%%%%%
\bibitem{DKMVZ} P. Deift, T. Kriecherbauer, K. T.-R. Mclaughlin, S. Venakides and X. Zhou, \textit{Asymptotics for Polynomials Orthogonal with Respect to Varying Exponential Weights}, Int. Math. Res. Not. (16), 1335--1425 (1997).
%%%%%
\bibitem{DVZ} P. Deift, S. Venakides and X. Zhou, \textit{New results in small dispersion KdV by an extension of the steepest descent method for Riemann-Hilbert problems}, Int. Math. Res. Not. (6), 285--299 (1997).
%%%%%
\bibitem{DZ} P. Deift and X. Zhou, \textit{A steepest descent method for oscillatory Riemann- Hilbert problems.
Asymptotics for the MKdV equation}, Ann. of Math \textbf{137}(2), 295--368 (1993).
%%%%%
\bibitem{DZNLS} P. Deift and X. Zhou, \textit{Long-time behavior of the non-focusing nonlinear Schr\"odinger equation – a case study}, New Series: Lectures in Math. Sciences Vol. 5, University of Tokyo, 1994.
%%%%%
\bibitem{dubr} B. Dubrovin, \textit{Theta functions and nonlinear equations}, Russian Math. Surveys \textbf{36}(2), 11--92 (1981).
%%%%%
\bibitem{EGKT} I. Egorova, Z. Gladka, V. Kotlyarov and G. Teschl \textit{Long-time asymptotics for the Korteweg--de Vries equation with step-like initial data},
Nonlinearity \textbf{26}(7), 1839--1864 (2013).
%%%%%
\bibitem{EGT} I. Egorova, Z. Gladka, and G. Teschl, {\em On the form of dispersive shock waves of the Korteweg--de Vries equation}, Zh. Mat. Fiz. Anal. Geom. \textbf{12}, 3--16 (2016).
%%%%%
\bibitem{EPT} I. Egorova, M. Piorkowski and G. Teschl, \textit{Asymptotics of the Korteweg--de Vries shock waves via the Riemann--Hilbert approach}, \arxiv{1907.09792}.
%%%%%
\bibitem{FK} H. Farkas and I. Kra, \textit{Riemann Surfaces}, $2^{\text{nd}}$ edition, Grad. Texts Math. \textbf{71}, Springer, New York, 1992.
%%%%%
\bibitem{GGM} M. Girotti, T. Grava, R. Jenkins and T.-R. McLaughlin, {\em Rigorous asymptotics of a KdV soliton gas}, Comm. Math. Phys. \textbf{384}, 733--784 (2021).
%%%%%
\bibitem{GT} K. Grunert and G. Teschl, \textit{Long-time asymptotics for the Korteweg--de Vries equation via nonlinear steepest descent}, Math. Phys. Anal. Geom. \textbf{12}(3), 287--324 (2009).
%%%%%
\bibitem{IM} A. Its and V. Matveev, \textit{Schr\"odinger operators with the finite-band spectrum and the
N-soliton solutions of the Korteweg–de Vries equation}, Teoret. Mat. Fiz. \textbf{23}(1), 51--68 (1975). (Russian)
%%%%%
\bibitem{KM10} V. Kotlyarov and A. Minakov, \textit{Riemann–Hilbert problem to the modified Korteveg–de Vries equation: Long-time dynamics of the steplike initial data}, J. Math. Phys. \textbf{51}, 093506 (2010).
%%%%%
\bibitem{KUK} S. B. Kuksin, {\em Analysis of Hamiltonian PDEs}, Oxford Lecture Series in Mathematics and its Applications {\bf 19}, Oxford University Press, Oxford, 2000.
%%%%%
\bibitem{JL} J. Lenells, \textit{Matrix Riemann-Hilbert problems with jumps across Carleson contours}, Monatsh. Math. \textbf{186}(1), 111--152 (2018).
%%%%%
\bibitem{MAR} V. A. Marchenko, {\em Sturm--Liouville Operators and Applications}, Birkh\"auser, Basel, 1986. 
\end{thebibliography}



\title[Riemann--Hilbert problem on the torus]{A scalar Riemann--Hilbert problem on the torus:
\\
Applications to the KdV equation}



\author[M. Piorkowski]{Mateusz Piorkowski}
\address{Department of Mathematics\\KU Leuven\\
Celestijnenlaan 200B 
\\
3001 Leuven}
\email{\href{mailto:Mathpiorkowski@gmail.com}{Mathpiorkowski@gmail.com}}

\author[G. Teschl]{Gerald Teschl}
\address{Faculty of Mathematics\\ University of Vienna\\
Oskar-Morgenstern-Platz 1\\ 1090 Wien}
\email{\href{mailto:Gerald.Teschl@univie.ac.at}{Gerald.Teschl@univie.ac.at}}
\urladdr{\href{http://www.mat.univie.ac.at/~gerald/}{http://www.mat.univie.ac.at/\string~gerald/}}


\keywords{Riemann--Hilbert problem, KdV equation, Jacobi theta functions}
\subjclass[2000]{Primary 	35Q15, 35Q53; Secondary 	30F10, 33E05}
\thanks{Research supported by the Austrian Science Fund (FWF) under Grants No.\ P31651 and W1245.}
\thanks{Anal. Math. Phys. {\bf 12}, 112 (2022)}

\begin{abstract}
We take a closer look at the Riemann--Hilbert problem associated to one-gap solutions of the Korteweg--de Vries equation. To gain more insight, we reformulate it as a scalar Riemann--Hilbert problem on the torus. This enables us to derive deductively the model vector-valued and singular matrix-valued solutions in terms of Jacobi theta functions. We compare our results with those obtained in recent literature.
\end{abstract}

\maketitle

%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Introduction}
\subsection{Background} The main goal of this short note is to present an alternative approach to the existence/uniqueness results for  the model Riemann--Hilbert (R-H) problem presented in \cite{EGKT} and the construction of a singular matrix-valued solution found in \cite[Sect.~6]{EPT} (see also \cite[Sect.~3]{GGM}). Recall, that the objective of \cite{EGKT} and \cite{EPT} was to rigorously apply the nonlinear steepest descent method to the initial value problem for the Korteweg--de Vries (KdV) equation,
\beq\nonumber
\partial_t q(x,t)=6q(x,t)\partial_x q(x,t)-\partial_x^3 q(x,t), \quad (x,t)\in\R\times\R_+,
\eeq
with steplike initial data $q(x,0) = q(x)$:
\beq\nonumber \lim_{x \to \infty} q(x) = 0, \qquad \lim_{x \to -\infty} q(x) = -c^2, \quad c > 0,
\eeq
satisfying certain smoothness conditions. For large $t$, solutions to this problem display  different behaviours in three regions of the $(x,t)$-plane characterized by the ratio $x/t$ (see \cite[Sect.~1]{EGKT}). Of particular interest to us is the transition region given by $-6c^2 t < x < 4c^2 t$, where solutions asymptotically converge to a modulated elliptic wave of the type \eqref{IM}. This result was proven in \cite{EPT}, where an ill-posedness of the corresponding holomorphic matrix model R-H problem was found.

The ill-posedness is closely related to the fact that the R-H problem for the KdV equation is formulated as a vector-valued problem. Note that, the standard Liouville-type argument relating existence to uniqueness for matrix-valued R-H problems having jump matrices with unit determinant (see for example \cite[Thm.~5.6]{JL}), cannot be generalized to the vector case in a straightforward manner. In fact, uniqueness can fail despite existence, as demonstrated in \cite[Sect.~2]{GT} for the simple case of a one-soliton solution. Uniqueness was restored by assuming the symmetry condition \eqref{symcond} below.

Another feature of the KdV equation playing an important role in the present note is the relationship between finite-gap solutions and elliptic Riemann surfaces (see \cite[Ch.~3]{BBEIM}). Algebro-geometric finite-gap solutions to the KdV equation can be given explicitly in terms of Jacobi theta functions via the Its--Matveev formula \cite{IM} (see also \cite{EGT}). Unsurprisingly, the solution of the corresponding model R-H problem is also expressed in terms of Jacobi theta functions. Given that these functions can be regarded as multivalued functions on an underlying Riemann surface, the natural question arises whether the model R-H problem in the plane found in \cite{EGKT} can be viewed as a R-H problem on a Riemann surface instead. In our simple one-gap case, that would correspond to a R-H problem on a torus. 

\subsection{Outline of this work}

In the next section we will recall the Its--Matveev formula for finite-gap solutions of the KdV equation together with the corresponding elliptic Riemann surface $\hat X$ and related quantities. 

In Section \ref{ModelRH} we state the one-gap KdV model R-H problem and show that it characterizes solutions to the KdV equation via its Lax pair representation. We then formulate the model R-H problem as a \emph{scalar-valued} R-H problem on another elliptic Riemann surface $X$, which can be viewed as a double covering of $\hat X$. Equivalently, solutions to this problem can be characterized by quasiperiodic meromorphic functions in the complex plane (see~\eqref{eqjcav}), leading to the explicit R-H model solution found in \cite{EGKT} and singular solutions similar to the one described in \cite{EPT}, \cite{GGM} in a straightforward manner. We show that in this picture the symmetry condition from \eqref{symcond} translates to halving the period (see \eqref{SymCond}), while uniqueness follows from Liouville's Theorem.

Section~\ref{sec:mv} discusses singular matrix-valued model solutions (see \cite{BKT16}, \cite{EPT}, \cite{GGM}). As shown in \cite{EPT}, there is no regular matrix-valued model solution satisfying all the standard assumptions, hence it is necessary to drop some of them. We compare our singular matrix-valued model solution to the ones found previously. We point out that the corresponding vanishing problem has a nontrivial solution, meaning that there is no uniqueness for the associated singular model problem. In particular, the solutions described in \cite{BKT16}, \cite{EPT} and \cite{GGM} differ from the one we present in Section~\ref{sec:mv}.



%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Riemann surfaces related to Its--Matveev solutions}\label{RiemannSurfaces}

The following section is based on the material found in \cite{EGT} (see also \cite{BBEIM} for more details). We will consider one gap solutions $q^{IM}(x,t)$ of Its--Matveev type, with spectrum given by
\begin{align*}
    \sigma := \sigma(H) = [-c^2, -a^2] \cup \R_+, \qquad 0 < a < c,
\end{align*}
where 
\begin{align*}
    H = -\partial_x^2 + q^{IM}, \qquad \dom(H) := H^2(\R) \subset L^2(\R),
\end{align*}
is the associated self-adjoint Schr\"odinger operator. To this end consider the elliptic Riemann surface $\hat X$ associated with the function
\[
 \mathcal{R}(\lambda) := \sqrt{\lambda (\lambda +c^2)(\lambda+a^2)},
 \]
with cuts along the spectrum $\sigma$. A point on $\hat X$ is denoted by $p = (\lambda,\pm)$, $\lambda \in \C \cup \lbrace \infty \rbrace$. 

Let $\lbrace {\bf\hat a}, \hatb\rbrace$ be a canonical basis on $\hat X$, where the cycle ${\hatb}$  surrounds the interval $[-c^2, -a^2]$ counterclockwise on the upper sheet and the cycle ${\bf\hat a}$ supplements ${\hatb}$ by passing along the gap $[-a^2, 0]$ in the positive direction on the lower sheet and then changing the sheet. We observe that $\hat X$ is a compact Riemann surface of genus one, and thus has, up to multiples, a unique holomorphic differential $d \hat \omega$, which can be normalized such that $\int_{\bf\hat a} d\hat \omega = 1$. We denote the integral over the $\hatb$ cycle by 
\begin{align*}
\hat \tau := \int_{\hatb} d\hat \omega. 
\end{align*}
Next introduce the Riemann constant $\hat K = -\frac{\hat \tau}{2} + \frac{1}{2}$ and define the wave and frequency numbers $V$ and $W$ (\cite{KUK}, \cite{MAR}) as the ${\hatb}$-periods of the normalized Abelian differentials of the second kind $d\hat \Omega_1$ and $d\hat \Omega_3$ on $\hat X$
uniquely defined by the order of the pole at infinity
\[
d\hat \Omega_1 = \frac{\I}{2\sqrt\la}(1+ O(\la^{-1}))d\la,
\hspace{10pt}
d\hat \Omega_3 = -\frac{3\I}{2}\sqrt\la (1 +O(\la^{-1}))d\la,\ \  \la\to\infty
,\] 
and by the normalization conditions
$
\int_{\bf \hat a} d \Omega_{1,3} = 0
$. To be precise, we set:
\begin{align}\label{defVW}
\I V := \int_{\hatb} d\hat \Omega_1, \qquad
\I W := \int_{\hatb} d\hat \Omega_3.
\end{align}
The explicit form of $d\hat \Omega_1$, which will be needed later, is given by (see \cite{EGT})
\begin{align}\label{Omega1hat}
    d\hat \Omega_1 := \frac{\I}{2}\frac{\lambda - h}{\mathcal R(\lambda)} d\lambda,
\end{align}
where $h := \int_{\bf \hat a} \frac{\lambda \, d\lambda}{\mathcal R(\lambda)}  \big( \int_{\bf \hat a} \frac{d\lambda}{\mathcal R(\lambda)} d\lambda \big)^{-1}$. Moreover, $d\hat\Omega_{1,3}$ change sign under the sheet exchange map $(\lambda,\pm) \to (\lambda, \pm)^* := (\lambda, \mp)$, see \cite[Eq.~2.3]{EGT}.

The final ingredient we will need is the theta function $\theta_3 = \theta_3( \, \cdot \, |\, \hat \tau)$ of $\hat X$ given by
\begin{align}\label{Theta}
    \theta_3(z\, | \, \hat \tau) := \sum_{n\in \Z} \exp\big\lbrace \pi\I\hat\tau n^2+2\pi\I z n\big\rbrace.
\end{align}
As $\I\hat \tau < 0$, this series is absolutely convergent for all $z \in \C$. We are now in the position to state the following 
\begin{theorem}[Its, Matveev \cite{IM}]
Let $\Delta \in \R$ be given. The function 
\begin{align}\label{IM}
    q^{IM}(x,t) := -2\frac{d^2}{dx^2} \log \theta_3\Big(\frac{1}{2\pi}(V x-4 W t+ \Delta)-\hat K \, \Big| \, \hat \tau \, \Big)-a^2-c^2-2h
\end{align}
defines a global solution of the KdV equation. Moreover, $\sigma(H) = [-c^2,-a^2]\cup \R_+$, where $H$ is the unique self-adjoint realization of the differential expression $-\partial^2_x+ q^{IM}$ in $L^2(\R)$. 
\end{theorem}
Similar formulas to \eqref{IM} exist for the higher genus case, see \cite[Sect.~3]{BBEIM}. We will show in the next section that the standard vector-valued R-H problem characterizing the solution \eqref{IM} can be formulated on an elliptic Riemann surface $X$, which will turn out to be a double covering of $\hat X$. The Riemann surface $X$ is more suitable for the nonlinear steepest descent analysis (see \cite{EGKT}, \cite{GGM}). 
%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{The model Riemann--Hilbert problem}
\label{ModelRH}
In the following we recall the model vector-valued R-H problem for one-gap solutions of the KdV equation. For the underlying scattering theory and nonlinear steepest descent analysis leading to this problem in the transition region, we refer to \cite[Sect.~4]{EGKT}.

We want to find a vector-valued  function $m^{\text{mod}}(k)=(m_1^{\text{mod}}(k),\ m_2^{\text{mod}}(k))$ holomorphic in the domain $\C\setminus [\I c,  -\I c]$, continuous up to the boundary except possibly at the points $\mathcal G:=\{ \I c, \I a, -\I a, -\I c\}$ and satisfying the jump condition:
\beq\label{defmvecmod}
m_+^{\text{mod}}(k)= m_-^{\text{mod}}(k) v^{\text{mod}}(k),
\eeq
(with $m_\pm(k):=\lim_{\eps\downarrow 0} m(k\pm\eps)$ as usual), where
\beq\label{jumpcondmod}
v^{\text{mod}}(k) = \left\{ \begin{array}{ll}
\begin{pmatrix}
0 & \I \\
\I  & 0
\end{pmatrix},& k\in [\I c, \I a], \,\\
\begin{pmatrix}
0 & -\I \\
-\I & 0
\end{pmatrix},& k\in [-\I a, -\I c],\\
\begin{pmatrix}\E^{-2\pi\I\ti{\Lambda}}& 0\\
0&\E^{2\pi\I\ti{\Lambda}}\end{pmatrix},& k\in [\I a, -\I a],\\
\end{array}\right.
\eeq
the symmetry condition
\beq \label{symcond}
m^{\text{mod}}(-k) = m^{\text{mod}}(k) \sigI,
\eeq
and the normalization condition
\beq\label{normcond}
\lim_{k\to\I\infty} m^{\text{mod}}(k)= (1\ \ 1).\eeq
At any point $\kappa\in\mathcal G$ the vector function $m^{\text{mod}}(k)$ can have at most a fourth root singularity:  $m^{\text{mod}}(k)= O((k-\kappa)^{-1/4})$, $k\to \kappa$.
\begin{figure}[H]
\begin{tikzpicture}
\draw[very thin] (-3,0) -- (3,0);
\draw[very thin] (0,-3) -- (0,3);
\draw[thick,->] (0,-1) -- (0,-1.75);\draw[thick] (0,-1.75) -- (0,-2.5);
\draw[thick,->] (0,2.5) -- (0,1.75);\draw[thick] (0,1.75) -- (0,1);
\draw[thick,dashed,->] (0,1.5) -- (0,0);\draw[thick,dashed] (0,0) -- (0,-1.5);
\draw[fill] (0,2.5) circle (0.03) node[above right] {$\I c$};
\draw[fill] (0,-2.5) circle (0.03) node[below right] {$-\I c$};
\draw[fill] (0,1) circle (0.03) node[above right] {$\I a$};
\draw[fill] (0,-1) circle (0.03) node[below right] {$-\I a$};

\node[above right] at (2.5,0) {$\mathbb{R}$};

\draw (0.2,0.2) node[above right,xshift=-3,yshift=-5] {$\big(\begin{smallmatrix} \E^{-2\pi\I\ti{\Lambda}}& 0\\ 0 & \E^{2\pi\I\ti{\Lambda}}\end{smallmatrix}\big)$};
\draw (-0.2,-1.5) node[below left] {$\big(\begin{smallmatrix} 0& -\I\\ -\I & 0\end{smallmatrix}\big)$};
\draw (-0.2,1.5) node[above left] {$\big(\begin{smallmatrix} 0& \I\\ \I & 0\end{smallmatrix}\big)$};

\end{tikzpicture}
\caption{Jump contour for the model R-H problem}
\end{figure}
In Theorem \ref{Theoremqm} we will see that the choice $\tilde \Lambda = \frac{1}{2\pi}(Vx-4Wt+\Delta)$ with $\Delta \in \R$, will lead to the KdV solution \eqref{IM}. 
\begin{remark}
The model problem presented here, together with its higher genus generalizations, appears frequently in applications of R-H theory as the outer parametrix problem, see for example \cite[Sect.~4.1]{BKT16}, \cite[Sect.~1]{DKMVZ}, \cite{DVZ}.
\end{remark}
Of crucial importance is the following fact about solutions of the model problem.
\begin{proposition}\label{PropUnique}
For any $\tilde \Lambda \in \R$, the model R-H problem has a unique solution. 
\end{proposition}
\begin{proof}
The proof can be found in \cite[Sect.~5]{EPT}. However, uniqueness will follow from the material presented later in this section, see in particular Lemma~\ref{uniqueLemma}.
\end{proof}
\subsection{Elliptic Riemann surface related to the model problem}
As a preparation for the proof of Theorem~\ref{Theoremqm} on the relation between the model R-H problem and the KdV equation, we have to consider the elliptic Riemann surface $X$ associated with the function
\[
w(k) :=\sqrt{(k^2 + c^2)(k^2 + a^2)},
\]
defined on $\C \setminus ([-\I c, -\I a] \cup [\I a, \I c])$ with the branch fixed by $w(0) > 0$. The two sheets of $X$ are glued along the cuts  $[\I c,\I a]$ and $[-\I a, -\I c]$. Points on this surface are denoted by $p=(k,\pm )$. To simplify formulas we keep the notation  $k=(k,+)$ for points on the upper sheet of $X$. Clearly $w$ extends to a meromorphic function on $X$ via $w(p)= w(k)$ for $p=(k,+)$ and $w(p)= -w(k)$ for $p=(k,-)$.

The Riemann surface $X$ is seen to be a double
covering of the Riemann surface $\hat X$ introduced in the previous section through the mapping
\begin{align}\label{CovMap}
    \Cov \colon X \to \hat X, \quad (k, \pm) \mapsto (k^2, \pm \sgn(\im k)),
\end{align}
where $\lbrace (k,\pm) : \im k = 0 \rbrace$ gets mapped to the cut $\R_+$ on $\hat X$. Alternatively, $\hat X \simeq X / \mathord{\sim}$, where $(k, +) \sim (-k,-)$. We can push forward the meromorphic differentials $d\hat \Omega_{1,3}$ by $\Cov$ to obtain meromorphic differentials $d\hat \Omega_{1,3}$ on $X$. In particular formula \eqref{Omega1hat} becomes
\begin{align}\label{Omega1Exp}
    d\Omega_1(k) = \I \frac{k^2-h}{w(k)} dk = (\I + O(k^{-2})) dk.
\end{align}

As in Section \ref{RiemannSurfaces} we choose a canonical homology basis of cycles $\{\bf a, \bf b\}$ for $X$ as follows: The $\bf a$-cycle surrounds the points $-\I a,\I a$ starting on the upper sheet from the left side of
the cut $[\I c,\I a]$ and continues on the upper sheet to the left part of $[-\I a, -\I c]$ and returns after changing sheets. The cycle $\bf b$ surrounds the points $\I a, \I c$ counterclockwise on the upper sheet. Note that the $\bf b$-cycle is mapped via $\Cov$ to the $\hatb$-cycle on $\hat X$, but the image of the $\bf a$-cycle encircles the $\bf \hat a$-cycle twice. Thus, the relation to the wave number and frequency remains unchanged,
\begin{align*}
    \I V = \int_{\bf b} d\Omega_1, \hspace{20pt} 
\I W = \int_{\bf  b} d\Omega_3,
\end{align*}
while the condition $\int_{\bf \hat a} d \Omega_{1,3} = 0$ now reads:
\begin{align*}
    \int_{-\I a}^{\I a} d\Omega_{1,3} = 0.
\end{align*}

Next we introduce the $g$-function, given by (c.f.~\cite[Eq.~2.17]{EGT})
\begin{align*}
    g(k) := \int_{\I c}^k 4\I \left( d\Omega_3 - \frac{\I x}{t} d\Omega_1 \right),
\end{align*}
where we restrict the path of integration to $\C \setminus [-\I c, \I c]$ on the upper sheet. Note that this $g$-function is different than the one presented in \cite[Lem.~4.1]{EGKT} in the steplike case, as it would lead to a modulated Its--Matveev solution. The following lemma holds.
\begin{lemma}
The function $g$ satisfies the following properties:
\begin{enumerate}
    \item $g(k)$ is an analytic function for $k \in \C \setminus [-\I c, \I c]$;
    \item  $g(k)=-g(-k)$ for $k\in \C\setminus [\I c, - \I c]$;
    \item $g_-(k)+g_+(k)=0$ as $k\in [\I c, \I a]\cup [-\I a, - \I c]$;
\item $t(g_-(k) - g_+(k))=Vx-4Wt$ as $k\in [\I a, - \I a]$;
\item the asymptotical behavior
\begin{align}\label{asymg}
 \Phi(k)-\I g(k)= O\left(\frac{1}{k}\right)
\end{align}
holds as $k\to\infty$, where we define the KdV-phase function to be
 \begin{align*}
     \Phi(k) = \Phi(k,x,t) := 4\I k^3 + \I k \frac{x}{t}. 
 \end{align*}
\end{enumerate}
\end{lemma}
Additionally, for Theorem \ref{Theoremqm} we will need the $x$-derivative of the $O(k^{-1})$-term in \eqref{asymg}.
\begin{lemma}
We have
\begin{align}\label{omega1}
    \int_{\I c}^k d\Omega_1 = \I k -\frac{1}{2\I k}(2h+a^2+c^2) + O(k^{-3}).
\end{align}
\end{lemma}
\begin{proof}
Computing the $O(k^{-2})$-term in \eqref{Omega1Exp} leads to
\begin{align*}
    d\Omega_1 = \big(\I + \frac{1}{2\I k^2}(2h+a^2+c^2) + O(k^{-4})\big) dk
\end{align*}
and hence
\begin{align*}
    \int_{\I c}^k d\Omega_1 = \I k + C -\frac{1}{2\I k}(2h+a^2+c^2) + O(k^{-3}).
\end{align*}
Since $g(k)$ is odd we must have $C=0$.
\end{proof}
In particular we have
\begin{align*}
t\Phi(k)-t\I g(k)= \frac{d(x,t)}{2k\I} + O(k^{-3})
\end{align*}
with 
\begin{align*}
    \partial_x d(x,t) = 2h+a^2+c^2.
\end{align*}
The importance of the model R-H problem lies in the following
\begin{theorem}\label{Theoremqm}
Assume $\tilde \Lambda = \frac{1}{2\pi}(Vx-4Wt+\Delta)$, $(x,t) \in \R^2$, where $V$ and $W$ are the wave number and frequency defined in \eqref{defVW} and $\Delta$ is an arbitrary real constant. Let $m^{\emph{mod}}(k,x,t)$ be the unique solution of the corresponding R-H problem, which we assume to be smooth in the variables $x$ and $t$. Define
\begin{align*}
    q^{\emph{mod}}(x,t) = \partial_x Q(x,t)-2h-a^2-c^2
\end{align*}
where
\begin{align*}
    m^{\emph{mod}}(k,x,t) = (1 \ \ 1) + \frac{Q(x,t)}{2k\I}(-1 \ \ 1) + O(k^{-2}).
\end{align*}
Then $q^{\text{mod}}(x,t)$ is a solution of the KdV equation.  
\end{theorem}
\begin{proof}
First we will introduce the vector-valued function 
\begin{align*}
    \Psi(k,x,t) = \Psi(k) = m^{\text{mod}}(k) \E^{-\I tg(k)\sigma_3},
\end{align*}
where $\sigma_3 = \begin{pmatrix}
1 & 0
\\
0 & -1
\end{pmatrix}$ and thus $\E^{-\I tg(k)\sigma_3} = \begin{pmatrix}
\E^{-\I tg(k)} & 0
\\
0 & \E^{\I tg(k)}
\end{pmatrix}$. 
Then $\Psi$ will satisfy the jump condition 
\begin{align}\label{jumpPsi}
    \Psi_+(k) = \Psi_-(k) v^\Delta(k),
\end{align}
where 
\begin{align*}
    v^{\Delta}(k) = \left\lbrace \begin{array}{ll}
\begin{pmatrix}
0 & \I \\
\I  & 0
\end{pmatrix},& k\in [\I c, \I a], \,\\
\begin{pmatrix}
0 & -\I \\
-\I & 0
\end{pmatrix},& k\in [-\I a, -\I c],\\
\begin{pmatrix}\E^{-\I \Delta}& 0\\
0&\E^{\I \Delta}\end{pmatrix},& k\in [\I a, -\I a],\\
\end{array} \right.
\end{align*}
the symmetry condition
\begin{align}\label{symPsi}
    \Psi(-k) = \Psi(k) \begin{pmatrix}
    0 & 1
    \\
    1 & 0
    \end{pmatrix},
\end{align}
the normalization condition
\begin{align*}
    \Psi(k) = \big[(1 \ \ 1) + \frac{Q-d}{2k\I}(-1 \ \ 1) + O(k^{-2})\big]   \E^{-t\Phi(k)\sigma_3}, 
\end{align*}
as $k \to \infty$, and has at most fourth root singularities for $k \to \kappa \in \mathcal G$.

Next introduce the Lax pair $L, P$ given by
\begin{align*}
    L = -\partial_x^2 + V, \qquad P = -4\partial_x^3+ 6V\partial_x + 3(\partial_x V)
\end{align*}
for some smooth potential $V = V(x,t)$ to be determined. Note that $L\Psi-k^2\Psi$ satisfies the jump condition \eqref{jumpPsi}, as the jump matrices are constant, and the symmetry condition \eqref{symPsi}. For $k\to\infty$ one obtains
\begin{align*}
    L\Psi(k)-k^2\Psi(k) = \big[\big(V-\partial_x(Q-d)\big) (1 \ \ 1) + O(k^{-1})\big] \E^{-t\Phi(k) \sigma_3}.
\end{align*}
Choosing 
\begin{align}\label{potential}
    V := \partial_x(Q-d),
\end{align}
we see that $L\Psi(k)-k^2\Psi(k) = O(k^{-1}) \E^{-t\Phi(k) \sigma_3}$, implying that $m_0(k) := (L\Psi(k)-k^2\Phi(k))\E^{\I tg(k)\sigma_3}$ would be a symmetric vanishing solution of the model problem. However, as the model problem has a unique symmetric solution by Proposition~\ref{PropUnique}, we can conclude $m_0 \equiv 0$ implying
\begin{align}\label{firstLaxEq}
    L\Psi = k^2\Psi.
\end{align}
Moreover, \eqref{firstLaxEq} implies that $\Psi$ has the form
\begin{align}\label{2ndorderExpansion}
    \Psi(k) = \big[(1 \ \ 1) + \frac{Q-d}{2k\I}(-1 \ \ 1) + \frac{R}{(2k\I)^2}(1 \ \ 1)+O(k^{-3})\big]   \E^{-t\Phi(k)\sigma_3},
\end{align}
where $\partial_x R = [\partial_x(Q-d)](Q-d)+\partial_{xx}(Q-d)$. 

Using \eqref{potential} and \eqref{2ndorderExpansion} we see, as before, that $(P\Psi - \partial_t \Psi)\E^{\I tg(k)\sigma_3}$ is a symmetric vanishing solution of the model problem implying
\begin{align*}
    P\Psi = \partial_t \Psi. 
\end{align*}
Now a standard computation shows that
\begin{align*}
    \underbrace{\big(\partial_t L - [P,L] \big)}_{\partial_t V - 6V\partial_x V + \partial_{xxx} V}\Psi = 0.
\end{align*}
As $\Psi$ is nonvanishing, this implies that $V = \partial_x(Q-d) = \partial_x Q -2h-a^2-c^2$ is a solution to the KdV equation. 
\end{proof}

\begin{remark}
It has been proven in \cite[Thm.~2.4]{EGT} that $q^{\emph{mod}} = \partial_x Q -2h-a^2-c^2$ is equal to the Its--Matveev solution \eqref{IM}.
\end{remark}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Construction of the solution to the model problem}
The solution to the model R-H problem for $m^{\text{mod}}(k)$ was given in \cite[Sect.~5]{EGKT}. As mentioned in the introduction, we want to solve this problem in a different way, which should shed some further light on the model problem. For this, it will be convenient to denote by $m(k)$ a generic vector-valued meromorphic function satisfying the jump condition \eqref{jumpcondmod}.

For our first transformation we define
\beq\nonumber
\ti{\ga}(k) = \sqrt[4]{\frac{k^2+a^2}{k^2+c^2}}
\eeq
with the branch cuts along $[\pm\I a, \pm\I c]$ and the branch chosen such that $\ti{\ga}(k)>0$ for $k\in [\I c, \I\infty)$.
Note that we have $\ti{\ga}(-k)=\ti{\ga}(k)$ and $\ti{\ga}(k)>0$ for $k\in\R$.
Then $\ti{\ga}(k)$ solves the scalar R-H problem
\begin{align*}
\ti{\ga}_+(k) &= \pm\I \ti{\ga}_-(k), \qquad k\in[\pm\I a, \pm\I c],
\\
\lim_{k\to \infty} \ti{\ga}(k) &= 1,
\end{align*}
and we set
\beq \label{mn}
m(k) = \ti{\ga}(k) n(k)
\eeq
such that $n(k)$ satisfies the jump condition \eqref{jumpcondmod}, except that the jumps on $[-\I c,-\I a]$ and $[\I c,\I a]$ are replaced by
\beq\label{eqjumpsigI}
n_+(k) = n_-(k) \sigI.
\eeq
In this setup, the two components $n_1$, $n_2$ of the vector $n \colon \C\setminus[-\I c,\I c] \to \C^2$ can be regarded as the values of a single function $N \colon X \to \C$ on the upper, lower sheet, respectively. Explicitly,
\beq \label{nN}
n(k) = (N((k,+)),N((k,-))).
\eeq
In this case the jump condition \eqref{eqjumpsigI} implies that $N$ will have no jump along the cuts, where the two sheets are glued together. However, the other jump
will remain. In fact, the jump contour on $X$ is a closed loop passing through the two branch points $-\I a$ and $\I a$ (from $\I a$ along the upper sheet to $-\I a$ and back to $\I a$ on the lower sheet), on which we have the jump condition
\beq\label{jumpN}
N_+(p) = N_-(p) \E^{-2\pi\I\ti{\Lambda}}.
\eeq
Note that the symmetry condition \eqref{symcond} translates to
\beq\label{symN}
N(p^*) = N(-p),
\eeq
where $(k,\pm)^* = (k,\mp)$ denotes the sheet exchange map and $-(k,\pm)=(-k,\pm)$.

The normalized holomorphic differential is given by
\beq\nonumber
d\om=\Gamma\frac{d\zeta}{w(\zeta)}, \quad \text{where}\ \Gamma:=\left(\int_{\bf a} \frac{d \zeta}{w(\zeta)}\right)^{-1}\in \I \R_-,
\eeq
such that $\int_{\bf a}d\om=1$ and 
\beq\nonumber
\tau=\int_{\bf b} d\om\in \I \R_+.
\eeq
One can check that $\hat \tau = 2\tau$. 

In the following we will make use of certain algebraic properties of $\theta_3$ introduced in Eq.~\eqref{Theta}, see for example \cite{Byrd}. Recall that $\theta_3$ is even, $\theta_3(-z\,\big|\,\tau)=\theta_3(z\,\big|\,\tau)$, and satisfies
\beq\label{jthper}
\theta_3(z+ n + \tau\ell\,\big|\,\tau)=\theta_3(z\,\big|\,\tau )\E^{- \pi\I \tau \ell^2 - 2\pi\I\ell z} \quad\text{for}\quad \ell,n \in \Z.
\eeq
Furthermore, let $A(p)=\int_{\I c}^p d\om$ be the Abel map \cite[Sect.~III.6]{FK} on $X$. We identify the upper sheet of $X$ with the complex plane $\C\setminus([\I c, \I a]\cup [-\I a, -\I c])$.
Restricting the path of integration to $\C \setminus [\I c, -\I c]$ we observe that $A(k)$ is a holomorphic function in that given domain with the following properties:
\begin{itemize}
\item $A_+(k)=-A_-(k) \pmod{1}$, \ for $k\in [\I c, \I a]\cup [-\I a, -\I c]$;
\item $A_+(k)-A_-(k)=-\tau$, \ for $k\in [\I a, -\I a]$;
\item $A(-k)=-A(k) + \frac{1}{2}$, \ for $k\in\C\setminus [\I c, -\I c]$, 
\item $A_+(\I a)=-\frac{\tau}{2} =-A_-(\I a)$, $A_+(-\I a) = -\frac{\tau}{2} + \frac{1}{2}$,  $A_-(-\I a)=\frac{\tau}{2} + \frac{1}{2}$.
\item $A(\infty)=\frac{1}{4}$, $A(k)= \frac{1}{4} -\Gamma k^{-1} + O(k^{-3})$, \ as $k\to\infty$.
\end{itemize}
For points on the lower sheet we set $A(p^*)=-A(p)$.
Finally, denote by $K= \frac{1+\tau}{2}$ the Riemann constant associated with $X$ and abbreviate $\infty_\pm= (\infty,\pm)$, $0_\pm= (0,\pm)$. Note that
$A(0_+)=\frac{1}{4}+\frac{\tau}{2}$. By Riemann's vanishing theorem \cite[Sect.~VI.3]{FK} the zeros of $\theta_3$ are simple and given by $z = K +\Z+\tau \Z$.

According to the Jacobi inversion theorem \cite[Sect.~III.6]{FK}, the Abel map $A$ maps our Riemann surface $X$ bijectively to its associated Jacobi variety $\C/ (\Z+\tau\Z)$ depicted in Figure~\ref{figjv}.
\begin{figure}[ht]
\begin{tikzpicture}
\path [fill=gray!40] plot (0,-1.5) -- (2,-1.5) -- (2,1.5) -- (0,1.5) --cycle;
\path [fill=gray!20] plot (-2,-1.5) -- (0,-1.5) -- (0,1.5) -- (-2,1.5) --cycle;
\draw (-4.5,3) -- (4.5,3);
\draw (-4.5,0) -- (4.5,0);
\draw (-4.5,-3) -- (4.5,-3);
\draw (-4,-3.5) -- (-4,3.5);
\draw (0,-3.5) -- (0,3.5);
\draw (4,-3.5) -- (4,3.5);
\draw[dashed] (-4.5,1.5) -- (4.5,1.5);
\draw[dashed] (-4.5,-1.5) -- (4.5,-1.5);

\draw (4,0) node[above right] {$1$};
\draw (0,3) node[above right] {$\tau$};

\draw[fill] (0,0) circle (0.03) node[above right] {$\I c$};
\draw[fill] (-2,0) circle (0.03) node[above left] {$-\I c$};
\draw[fill] (2,0) circle (0.03) node[above right] {$-\I c$};
\draw[fill] (-1,0) circle (0.03) node[below] {$\infty_-$};
\draw[fill] (1,1.5) circle (0.03) node[below] {$0_+$};
\draw[fill] (-1,1.5) circle (0.03) node[below] {$0_-$};
\draw[fill] (1,-1.5) circle (0.03) node[below] {$0_+$};
\draw[fill] (-1,-1.5) circle (0.03) node[below] {$0_-$};
\draw[fill] (1,0) circle (0.03) node[below] {$\infty_+$};
\draw[fill] (0,1.5) circle (0.03) node[above right] {$\I a$};
\draw[fill] (0,-1.5) circle (0.03) node[below right] {$\I a$};
\draw[fill] (-2,1.5) circle (0.03) node[above] {$-\I a$};
\draw[fill] (2,1.5) circle (0.03) node[above] {$-\I a$};
\draw[fill] (-2,-1.5) circle (0.03) node[below] {$-\I a$};
\draw[fill] (2,-1.5) circle (0.03) node[below] {$-\I a$};

\end{tikzpicture}
\caption{Images of important points on the Jacobi variety (dark/light gray denotes the upper/lower sheet)}\label{figjv}
\end{figure}
The jump contour is indicated by the dashed line, while the dark/light shaded region correspond to the upper/lower sheet. Moreover, 
if we set
\beq \label{NE}
E(A(p))=N(p)
\eeq
for some meromorphic function $E(z)$, then $N$ will satisfy our original jump condition \eqref{jumpN} if and only if
\beq\label{eqjcav}
E(z+1) = E(z), \qquad E(z+\tau) =  E(z)\E^{2\pi\I\ti{\Lambda}},
\eeq
and it will satisfy the symmetry condition if and only if
\beq \label{SymCond}
E(z+\tfrac{1}{2}) = E(z).
\eeq
If this latter condition holds we will call $E$ symmetric. If we have $E(z+\frac{1}{2}) = -E(z)$, we will call $E$ anti-symmetric.  

At this stage we remind the reader that we have four equivalent ways of describing vector-valued functions satisfying the jump condition \eqref{jumpcondmod}:
\begin{align*}
    m(k) \quad \longleftrightarrow \quad n(k) \quad \longleftrightarrow \quad N(p) \quad \longleftrightarrow \quad E(z)
\end{align*}
related via \eqref{mn}, \eqref{nN} and \eqref{NE} respectively.
The most convenient framework will be given through the quasiperiodic meromorphic functions $E(z)$ characterized by \eqref{eqjcav}. Let us consider the space $\FOm$ of all quasiperiodic meromorphic functions on $\C/ (\Z+\tau\Z)$ satisfying \eqref{eqjcav}, without imposing any symmetry requirements.
Note that $E\in\FOm$ is uniquely determined up to a constant by its divisor $(E)= \sum_{j=1}^n \mathcal{D}_{z_j}- \sum_{j=1}^n \mathcal{D}_{p_j}$, 
since the quotient of two such functions with the same divisor is elliptic (i.e., doubly periodic with periods $1$ and $\tau$) without poles, hence a constant. Moreover, since $\frac{E'}{E}$ is elliptic, integrating this function along a fundamental domain (i.e.\ a  parallelogram generated by $1$ and $\tau$)
shows that the number of zeros and poles must be equal. Note also that there must be at least one pole (unless $\ti{\Lambda}=0$). Integrating
$z\frac{E'(z)}{E(z)}$ along a fundamental domain gives
\beq\label{eqabelom}
\sum_{j=1}^n z_j - \sum_{j=1}^n p_j = \ti{\Lambda} \pmod{\Z+\tau\Z}.
\eeq
Choosing representatives $z_j$, $p_j \in \C$ such that
\beq\label{eqabelom2}
\sum_{j=1}^n z_j - \sum_{j=1}^n p_j = \ti{\Lambda} \pmod{\Z},
\eeq
we can represent $E$ as
\beq\label{eqFtheta}
E(z) = E_0\prod_{j=1}^n \frac{\theta_3(z-z_j-K|\tau)}{\theta_3(z-p_j-K|\tau)}.
\eeq
Indeed the right-hand side has the required zeros and poles while \eqref{jthper} and \eqref{eqabelom2} ensure that it satisfies the quasiperiodicity conditions \eqref{eqjcav}.


\begin{lemma}
The divisor of $E\in\FOm$ is invariant with respect to translations of $\frac{1}{2}$ if and only if $E$ is either symmetric or anti-symmetric
\end{lemma}

\begin{proof}
Observe that $C=\frac{E(z+\frac{1}{2})}{E(z)}$ is elliptic without poles and hence constant. Moreover, $C^2=\frac{E(z+\frac{1}{2})}{E(z)} \cdot \frac{E(z+1)}{E(z+\frac{1}{2})} =1$
shows $C=\pm 1$. The converse is trivial.
\end{proof}

\begin{lemma} \label{uniqueLemma}
If $E\in\FOm$ is (anti-)symmetric and has at most two poles, it is already uniquely determined up to a constant by its poles. Conversely,
for each choice of two poles $p_1$, $p_2=p_1+\frac{1}{2}$ there is a unique (up to constants) symmetric and a unique anti-symmetric
function $E\in\FOm$, with at most simple poles at $p_1$ and $p_2$. In fact $p_1$, $p_2$ are simple poles, unless $\tilde \Lambda \in \mathbb{Z}$, in which case the symmetric solution is constant.  
\end{lemma}

\begin{proof}
Let $E$ be (anti-)symmetric and nonconstant. Denote its poles by $p_1$, $p_2=p_1+\frac{1}{2} \pmod{\Z+\tau\Z}$ and its zeros by $z_1$, $z_2=z_1+\frac{1}{2} \pmod{\Z+\tau\Z}$.
Choosing representatives in $\C$, \eqref{eqabelom} implies $2(z_1- p_1) = \tilde \Lambda + \ell +n \tau$ for some $\ell,n\in\Z$.
In particular, since adding a period to $z_1$ is irrelevant, we can assume $\ell,n\in\{0,1\}$. If $\ell=1$ this just amounts to exchanging
$z_1$ and $z_2$ and hence we can assume $\ell=0$ without loss of generality. Now using
$z_1 = p_1 + \frac{\tilde \Lambda}{2} + n\frac{\tau}{2}$, we can set $z_2= z_1 + \frac{1}{2} - n \tau$ and $p_2 = p_1 + \frac{1}{2}$ such that \eqref{eqabelom2}  holds. Now one computes using \eqref{jthper} that \eqref{eqFtheta} fulfills $E(z+\frac{1}{2})=(-1)^n E(z)$. In other words, $p_1 \in \C/(\Z+\tau\Z)$ and $n \in \Z_2$ uniquely determine $E$ up to
a constant. One can check, that the zeros and poles cancel if and only if $n = 0$ and $\tilde \Lambda \in \mathbb{Z}$, corresponding to a constant symmetric solution.
\end{proof}

\begin{corollary}\label{corc}
If $E\in\FOm$ has at most two poles $p_1$, $p_2=p_1+\frac{1}{2}$ then there exist unique $c_s,c_a \in \C$ such that $E=c_s E_s + c_a E_a$,
where $E_s$, $E_a$ are the symmetric, anti-symmetric solutions constructed in the previous lemma, respectively.
\end {corollary}

Returning to our original model problem, we want the poles of $E$ to lie at the images of $\I a$ and $-\I a$ under the Abel map $A$, that is $p_1=\frac{\tau}{2}$ and $p_2=\frac{1+\tau}{2} = K$. The reason is that we require $m^{\text{mod}}(k)$ to be holomorphic, with at most fourth root singularities at points of $\mathcal G$. As $\ti \ga(k)$ has fourth root zeros at $\pm \I a$ and the Abel map $A$ has  square root singularities at the points of $\mathcal G$, simple poles at $\frac{\tau}{2}$, $\frac{1+\tau}{2}$ in $\C/(\Z+\tau\Z)$  translate to fourth root singularities at $\pm \I a$ of $m^{\text{mod}}(k)$ under the inverse of the Abel map. In fact, this is the only choice of the pole structure leading to a $m^{\text{mod}}(k)$ holomorphic away from the jump contour with at most fourth root singularities at the contour. 

For the zeros of the symmetric ($n = 0$) and the anti-symmetric ($n = 1$) solution we use $z_1 = \frac{\ti{\Lambda}}{2} + \frac{(n+1) \tau}{2}$, $z_2=  \frac{\ti{\Lambda}}{2} + \frac{1-(n-1)\tau}{2}$. 
Denote by
\begin{align*} %\label{SymSol}
E_s(z)= \frac{\theta_3(z-\frac{\ti{\Lambda}}{2}+\frac{1}{2}|\tau)\theta_3(z-\frac{\ti{\Lambda}}{2}|\tau)}{\theta_3(z+\frac{1}{2}|\tau)\theta_3(z|\tau)}
\end{align*}
the corresponding symmetric and by
\begin{align*}%\label{AntiSol}
E_a(z)= \frac{\theta_3(z-\frac{\ti{\Lambda}}{2}-\frac{1+\tau}{2}|\tau)\theta_3(z-\frac{\ti{\Lambda}}{2}+\frac{\tau}{2}|\tau)}{\theta_3(z+\frac{1}{2}|\tau)\theta_3(z|\tau)}
\end{align*}
the corresponding anti-symmetric solution. Using the identity (cf.~ \cite{dubr} formula (1.4.3))
\[
 \textstyle \theta_3(z|\tau) \theta_3(z+ \frac{1}{2} |\tau) = \theta_3(2z+\frac{1}{2} | 2\tau)\,\theta_3(\frac{1}{2} | 2\tau),
 \]
 (note that the quotient of both sides is a holomorphic elliptic function which equals $1$ at $z=\frac{1}{2}$) we can write the formula for $E_s$ somewhat more compactly as
 \begin{align}\label{twotau}
 E_s(z) = \frac{\theta_3(2z-\ti{\Lambda}+ \frac{1}{2}|2\tau)}{\theta_3(2z+\frac{1}{2}|2\tau)}.
 \end{align}
This is the analytic counterpart to the observation \eqref{SymCond}, which states that the symmetry condition is related to halving of the (real) period, or equivalently to doubling of the half-period ratio $\tau$. Indeed, $E_s$ can be viewed as a function on the Riemann surface obtained by just taking one copy of $\mathbb{C} \setminus ([-\I a, -\I c]\cup[\I c, \I a])$ and gluing the two cuts together. The resulting Riemann surface has period $2\tau = \hat \tau$ and is thus biholomorphically equivalent to $\hat X$ defined in Section \ref{RiemannSurfaces}. More explicitly we have that the covering map \eqref{CovMap}, viewed as a map between Jacobi varieties, takes on the simple form
\begin{align*}
    \Cov \colon X \simeq \C / (\Z + \tau \Z) \to \hat X \simeq \C / (\Z + \hat \tau \Z), \qquad z \mapsto 2z.
\end{align*}

We see that for the symmetric case $n=0$, we have $\I\im(z_j) = \frac{\tau}{2} \pmod{\Z+\tau\Z}$ and both zeros will be on $[-\I a, \I a]$ (see Figure \ref{figjv}).
In the anti-symmetric case $n=1$ we have $\I\im(z_j) =0 \pmod{\Z+\tau\Z}$ and both zeros will be on $(\infty,-\I c] \cup [\I c, \infty]$.
In particular, if $\ti{\Lambda}=\frac{1}{2} \pmod {1}$ the two zeros of the anti-symmetric solution will be at $\infty_\pm$ and we cannot normalize at this point.
Moreover, if $\ti{\Lambda}=0 \pmod {1}$ such that we are looking for elliptic functions, we have $E_s(z)=1$ (i.e.\ zeros and poles coincide) and
the zeros of $E_a(z)$ will be at $z_1=0$ and $z_2=\frac{1}{2}+\tau$.

From Corollary \ref{corc} it follows that all solutions of \eqref{jumpN} with poles at most at $\pm\I a$ are given by
\[
N(p) = c_s N_s(p) + c_a N_a(p), \quad N_s(p)=E_s(A(p)), \ N_a(p)=E_a(A(p)), \quad c_s,c_a \in \C,
\]
and we have
\[
N(\infty_\pm)= c_s E_s(\tfrac{1}{4}) \pm c_a E_a(\tfrac{1}{4}), \qquad
N(0_\pm)= c_s E_s(\tfrac{1}{4}+\tfrac{\tau}{2}) \pm c_a E_a(\tfrac{1}{4}+\tfrac{\tau}{2}),
\]
with $E_s(\frac{1}{4})\ne 0$ for all $\ti{\Lambda}\in\R$ and $E_a(\frac{1}{4})\ne 0$ for all $\ti{\Lambda}\neq \frac{1}{2} \pmod{1}$. Moreover, $N$ will satisfy
\eqref{symN} if and only if $c_a=0$.

Note that in the special case $\ti{\Lambda}=0 \pmod {1}$ we have (up to constants):
\[
N_s((k,\pm))=1, \qquad N_a((k,\pm))= \frac{k^2+ c^2}{\pm w(k)}.
\]
In the case $\ti{\Lambda}=\frac{1}{2} \pmod {1}$ we have (again up to constants):
\[
N_s((k,\pm))= \frac{k}{\sqrt{k^2+a^2}}, \qquad N_a((k,\pm))=  \frac{1}{\sqrt{k^2+a^2}},
\]
where the root has the branch cut along $[-\I a,\I a]$.

Returning to our original problem we have shown:

\begin{lemma}
The function
\beq\nonumber
m^{\emph{mod}}(k) = \frac{\ti{\ga}(k)}{N_s(\infty_+)} \big(N_s((k,+)),N_s((k,-))\big)
\eeq
is the unique vector-valued function which is holomorphic in the domain $\C\setminus [\I c,  -\I c]$, has square integrable boundary values,
and satisfies the jump condition \eqref{defmvecmod}, the symmetry condition \eqref{symcond} and the normalization condition \eqref{normcond}.

Specifically, $m^{\text{mod}}(k)$ is continuous up to the boundary except at points of the set
$\mathcal G:=\{ \I c, \I a, -\I a, -\I c\}$ where it has at most a fourth root singularity:  $m^{\text{mod}}(k)= O((k-\kappa)^{-1/4}))$, $k\to \kappa$.
\end{lemma}
As shown previously in \cite{EGKT} and \cite{EGT}, the $m^{\text{mod}}(k)$ thus defined gives rise via Theorem \ref{Theoremqm} to an Its--Matveev solution \eqref{IM} related to the elliptic Riemann surface $\hat X$. 
%%%%%%%%%%%%%%%%%%%%%%%
\section{Matrix-valued solutions}
\label{sec:mv}



In the framework of the nonlinear steepest descent analysis one usually needs to construct a matrix-valued R-H solution which is invertible. This is a necessary step to arrive at a small-norm R-H problem which can be solved via a Neumann series. However, while many integrable wave equations like the modified KdV equation \cite{DZ}, \cite{KM10} or the nonlinear Schr\"odinger equation \cite{DZNLS} have a matrix-valued R-H formulation, this is not the case for the KdV equation. 

Recall that the model matrix-valued R-H problem related to the KdV equation has a jump matrix satisfying (see \cite{EGKT}, \cite{GT})
\begin{align} \label{symcondV}
    v^{\text{mod}}(-k) = \sigma_1 (v^{\text{mod}}(k))^{-1} \sigma_1, \qquad \det v(k) = 1, \quad k \in [\I c, -\I c].
\end{align}
For the corresponding holomorphic matrix-valued solution $M^{\text{mod}}(k)$, one would then require
\begin{align} \label{normcondM}
    \lim_{k \to \infty} M^{\text{mod}}(k) = \begin{pmatrix}
    1 & 0
    \\
    0 & 1
    \end{pmatrix}.
\end{align}
Moreover, given holomorphicity of $M^{\text{mod}}(k)$, we can derive from \eqref{symcondV} and \eqref{normcondM}:
\begin{align} \label{detcondM}
    M^{\text{mod}}(-k) = \sigma_1 M^{\text{mod}}(k) \sigma_1, \qquad \det M^{\text{mod}}(k) \equiv 1.
\end{align}
Note that \eqref{detcondM} implies that $M^{\text{mod}}(k)$ must have the form
\begin{align*}
    M^{\text{mod}}(k) = \begin{pmatrix}
    \tilde \alpha(k) & \tilde \beta(-k)
    \\
    \tilde \beta(k) & \tilde \alpha(-k)
    \end{pmatrix}
\end{align*}
and that $(1, 1)M^{\text{mod}}(k)$ will satisfy the symmetry condition \eqref{symcond}.

As a first attempt of writing down an invertible matrix-valued model solution with at most fourth root singularities near the points $\kappa \in \mathcal G$ (c.f. Corollary \ref{corc}), we start with the function
\[
M^{\text{mod}}_1(k) = \ti{\ga}(k) \begin{pmatrix}\alpha_1(k) & \beta_1(-k)\\ \beta_1(k) & \alpha_1(-k)\end{pmatrix}
\]
where
\begin{align*}
\alpha_1(k)&= \frac{1}{2} \big( N_a(\infty_+) N_s(k) + N_s(\infty_+) N_a(k) \big), 
\\
\beta_1(k)&= \frac{1}{2} \big( N_a(\infty_+) N_s(k) - N_s(\infty_+) N_a(k) \big).
\end{align*}
Note that $M^{\text{mod}}_1(k)$ satisfies the symmetry condition in \eqref{detcondM} and satisfies the normalization \eqref{normcondM} up to a multiplicative factor. In fact we have
\begin{align*}
    \lim_{k\to\infty}M^{\text{mod}}_1(k) = N_s(\infty_+) N_a(\infty_+) \begin{pmatrix} 1 & 0\\ 0 & 1\end{pmatrix}, 
\end{align*}
with
\begin{align*}
    \det M^{\text{mod}}_1(k)= N_s(\infty_+)^2 N_a(\infty_+)^2.
\end{align*}
The problem here is that the prefactor vanishes for $\tilde \Lambda = \frac{1}{2}\pmod 1$ as $N_a(\infty_+) = 0$, hence $N_a(k)$ is a vanishing solution of the model problem for these values $\tilde \Lambda$. It follows that we cannot enforce the normalization \eqref{normcondM} for all $\tilde \Lambda$. In particular, $M^{\text{mod}}_1(k)$ is not invertible for these values of $\tilde \Lambda$. The relation to $m^{\text{mod}}(k)$ is given through
\begin{align*}
    m^{\text{mod}}(k) = \frac{1}{N_s(\infty_+) N_a(\infty_+)} (1, 1)  M^{\text{mod}}_1(k).
\end{align*}
where one needs to use the rule of l'H\^ospital for $\tilde \Lambda = \frac{1}{2} \pmod 1$. 


To obtain an invertible matrix-valued model solution also for $\tilde \Lambda = \frac{1}{2}\pmod 1$ we will have to give up on holomorphicity and introduce poles. To be precise, we will move the poles of the anti-symmetric solution to $\check p_{1,2} = \pm \frac{1}{4} + \frac{\tau}{2}$, which corresponds to a pole at $(0, \pm)$ on $X$. Following Section \ref{ModelRH}, this gives rise to an anti-symmetric solution of the form
\begin{align} \label{singE}
    \check E_a(z) = \frac{\theta_3(z+\frac{1}{4}-\frac{\ti \Lambda}{2}-\frac{\tau}{2}|\tau)\theta_3(z-\frac{1}{4}-\frac{\ti \Lambda}{2}+\frac{\tau}{2}|\tau)}{\theta_3(z+\frac{1}{4}|\tau)\theta_3(z-\frac{1}{4}|\tau)}, \qquad \check N_a(p) = \check E_a(A(p)).
\end{align}
We can now define $M^{\text{mod}}_2(k)$  analogously to $M^{\text{mod}}_1(k)$, but substituting $\check N_a(k)$ for $N_a(k)$, and including the correct normalization at infinity for $\ti \Lambda \not = 0 \pmod 1$:
\[
M^{\text{mod}}_{2}(k) = \frac{\ti{\ga}(k)}{N_s(\infty_+)\check N_a(\infty_+)} \begin{pmatrix}\alpha_2(k) & \beta_2(-k)\\ \beta_2(k) & \alpha_2(-k)\end{pmatrix}, \qquad \ti \Lambda \not = 0 \pmod 1,
\]
where
\begin{align*}
\alpha_2(k)&= \frac{1}{2} \big(\check N_a(\infty_+) N_s(k) + N_s(\infty_+) \check N_a(k) \big), 
\\
\beta_2(k)&= \frac{1}{2} \big( \check N_a(\infty_+) N_s(k) - N_s(\infty_+) \check N_a(k) \big),
\end{align*}
Note that $N_s(\infty_+)\check N_a(\infty_+)\not = 0$ for $\ti \Lambda \not = 0 \pmod 1$ and thus we have:
\begin{align*}
    \lim_{k\to \infty} M^{\text{mod}}_2(k) = \begin{pmatrix}
    1 & 0
    \\
    0 & 1
    \end{pmatrix}, \qquad \ti \Lambda \not = 0 \pmod 1.
\end{align*}
Moreover, $\det M^{\text{mod}}_2(k)$ is an even meromorphic function with at most a simple pole at the origin, hence $\det M^{\text{mod}}_2(k) \equiv \det M^{\text{mod}}_2(\infty)= 1$. We do not define $M^{\text{mod}}_2(k)$ for $\ti \Lambda = 0 \pmod 1$, which should not pose a problem in applications as the standard holomorphic matrix-valued solution $M^{\text{mod}}_1(k)$ can be constructed in this case.

It turns out that sacrificing holomorphicity, while retaining \eqref{normcondM} and \eqref{detcondM}, is the most convenient way to deal with the ill-posedness of the holomorphic matrix-valued model problem for the KdV equation. 
Indeed, this was the strategy in \cite[Sect.~6]{EPT} and \cite[Sect.~3.3]{GGM}. Note however, that the anti-symmetric meromorphic vector solutions found \cite{EPT} and \cite{GGM} are not the same as given by \eqref{singE}. The reason is, that while we assumed that $\check N_a(p)=\check E_a(A(p))$ has only poles at $0_\pm$, this condition was more strict than necessary, as we could still allow for singularities at $\pm \I a$, as is the case for $m^{\text{mod}}(k)$. Indeed, $m^{\text{mod}}(k)/k$ has a pole at the origin and is an anti-symmetric solution to the vanishing problem where solutions are required to vanish at infinity. Hence, there is no chance for uniqueness if we allow for poles at $0$ and fourth root singularities at $\pm \I a$. Moreover for $\ti \Lambda = 0 \pmod 1$, $m^{\text{mod}}(k)/k$ has no singularities at $\pm \I a$, and hence by our uniqueness Lemma \ref{uniqueLemma} must coincide with the solution generated by $\check N_a(p)$. 

Interestingly, any anti-symmetric solution with a simple pole at the origin and fourth root singularities at $\pm \I a$, which is normalized to $(-1 \ \ 1)$ at infinity, is adequate for the analysis performed in \cite{EPT}. The reason is that the pole cancellation in the final step of the nonlinear steepest descent analysis is due to the underlying symmetry class, rather than the exact form of the second vector-valued solution (see Lemma 6.4 in \cite{EPT}).  

On a more geometric side, it is interesting to observe that while the model R-H solution $m^\text{mod}(k)$ can be conveniently defined on a 'smaller' Riemann surface $\hat X$ as in \eqref{twotau}, for the nonlinear steepest descent analysis it seems to be necessary to consider the Riemann surface $X$ which is a double covering of $\hat X$.
\\
\\
%\noindent{\bf Data Availability Statement.} Data sharing not applicable to this article as no datasets were generated or analysed.


%%% Partial Proof of Lemma 3.3
\begin{comment}
\textbf{Maybe we can skip the proof, seems elementary.}

Note that while the $\hatb$-cycle on $\hat X$ is mapped to the $\bf b$-cycle on $X$, the $\hatb$-cycle on $\hat X$ gets mapped to a path from $-\I a$ to $\I a$. Hence the conditions that the $\bf \hat a$-cycles of $d\Omega_{1,3}$ vanish, implies that
\begin{align*}
    \int_{-\I a}^{\I a} d\Omega_{1,3} = \int_{-\I c}^{\I c} d\Omega_{1,3} = 0. 
\end{align*}
This shows that $g$ is a holomorphic function in $\C \setminus [-\I c, \I c]$. It now follows that
\begin{align*}
    g(-k) = \int_{\I c}^{-k} 4\I d\Omega_3 - \frac{\I x}{t} d\Omega_1 &= \int_{-\I c}^{-k} 4\I d\Omega_3 - \frac{\I x}{t} d\Omega_1 
    \\
    &= -\int_{\I c}^{k} 4\I d\Omega_3 - \frac{\I x}{t} d\Omega_1 = -g(k)
\end{align*}
This follows from the fact that $d\Omega_{1,3}$ differ by a sign at the points $p_\pm = (\lambda, \pm) \in \hat X$. Hence, we conclude that
\begin{align*}
    g(k) = \int_{\I c}^k 4\I d\Omega_3 - \frac{\I x}{t} d\Omega_1 = 4k^3 -\frac{\I x}{t} k + O(k^{-1}), \qquad k \to \infty
\end{align*}
or equivalently
\begin{align}\label{asymg}
    \Phi(k) - \I g(k) = O(k^{-1}), \qquad k \to \infty.
\end{align}
\textbf{End of partial proof}
\end{comment}
\end{document}